\documentclass[12pt]{article}
\setlength{\parindent}{0in}
\setlength{\parskip}{0.08in}
\setlength{\topmargin}{-0.5in}
\addtolength{\textheight}{1.2in}
\pagestyle{empty}
\begin{document}

\begin{center}
{\large \textbf{Mathematics 468}} \\
\textbf{Topology} \\
\textbf{Spring 1999}
\end{center}
\vspace{-0.05in}

\begin{tabbing}
\textsc{Lecture time and place:} MWF 10:40--11:30, Nieuwland 182 \\
\textsc{Instructor:} \= John Palmieri \\
                   \> 205 CCMB, 631-8846 \\ 
                   \> E-mail: \texttt{John.H.Palmieri.2@nd.edu} \\
\textsc{Office hours:} \= Monday 9:30--10:30, 3:00--4:00 \\
 		       \> (or just drop by---I'm around a lot) \\
\textsc{Web:} \texttt{http://www.nd.edu/\~{}jpalmier/Math468/} \\
\textsc{Text:} \emph{A Combinatorial Introduction to Topology} by
M.~Henle
\end{tabbing}
\vspace{-0.15in}

\textsc{Examinations and Grading:} There will be a one-hour exam and a
final; there will also be regular homework assignments.  The one-hour
exam is worth 100 points, the final is worth 150 points, and the
homework will count for 150 points, for a total of 400 points.

\textsc{Hour exam:} date to be announced

\textsc{Final exam:} 8:00--10:00 am, Wednesday, May 5, location to be
announced.

\textsc{Homework:} Homework will be assigned frequently.  Feel free to
work with other people on your homework, but you must write your
solutions yourself.  Some class time will be spent going over homework
problems; I will ask each of you to present solutions on the
blackboard.

\textsc{Honor code:} Exams and homework are conducted under the honor
code.  As mentioned above, cooperation on homework is permitted, but
copying is not.

\textsc{Plan for the course:} We will follow the book pretty closely.

\end{document}

